Computability, Unsolvability, Randomness Math 497A: Collaboration Policy

Stephen G. Simpson

First draft October 8, 2007 This draft October 8, 2007

1. Students are allowed/encouraged to work together and help/consult with each other on homework assignments, studying, etc. Such collaboration is encouraged to the extent that it may lead to improved individual understanding.

2. Each student is required to write up his/her homework individually in his/her own words reflecting his/her own detailed individual understanding. Copying from the papers of other students is not permitted.